Author’s response to reviews

Title: Rapidly progressing, fatal acute promyelocytic leukaemia that initially manifested as a painful third molar: a case report

Authors:

Juan A Suárez-Cuenca (jsuarez@ifc.unam.mx)
José L Arellano-Sánchez (md_arlos79@hotmail.com)
Aldo A Scherling-Ocampo (aldosterona1@hotmail.com)
Gerardo Sánchez-Hernández (gmtp@prodigy.net.mx)
David Pérez-Guevara (dvprz@hotmail.com)
Juan R Chalapud-Revelo (injuan@hotmail.com)

Version: 8 Date: 31 October 2009

Author’s response to reviews:

Dear JMCR editorial team:

As requested, we reviewed the edited version, and only minor changes were made in: author’s institutions and format of table 1. Now, the manuscript is ready for publication.

Kind regards.